Author’s response to reviews

Title: Validation of the Pelvic Organ Prolapse Simple Screening Inventory (POPSSI) in a population of Ethiopian women.

Authors:

Dawit kassa (davworku@yahoo.com)
Yirgu Ferede (yirgug@yahoo.com)
Polina Advolodkina (padvolo@emory.com)

Version: 3 Date: 04 Mar 2019

Author’s response to reviews:

Dear Editors, thank you for the continued support. we have edited the manuscript and incorporated all the comments given and followed the journals submission guideline. The Declaration section have also been moved just after the conclusion section. Regarding the 3rd question “Please clarify whether the questionnaire used in your study was developed for this study or has previously been published elsewhere”: the questionnaire have been published in a journal that is mentioned as reference number 9.